

**f** 

# Water Resources Research<sup>®</sup>

# **RESEARCH ARTICLE**

10.1029/2022WR033734

#### **Key Points:**

- Statistical models accurately predict summer meteorological and agricultural drought at 1-, 2-, and 3-month lead times at a 4-km resolution
- Most of the interannual variability in summer drought area can be explained by preceding cold-season climate conditions
- Statistical models rely heavily on the autocorrelation of drought indices in drought forecasts

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

R. Abolafia-Rosenzweig, abolafia@ucar.edu

#### Citation:

Abolafia-Rosenzweig, R., He, C., Chen, F., Ikeda, K., Schneider, T., & Rasmussen, R. (2023). High resolution forecasting of summer drought in the western United States. *Water Resources Research*, *59*, e2022WR033734. https:// doi.org/10.1029/2022WR033734

Received 21 SEP 2022 Accepted 6 MAR 2023

#### **Author Contributions:**

Conceptualization: Ronnie Abolafia-Rosenzweig, Cenlin He Data curation: Kyoko Ikeda Formal analysis: Ronnie Abolafia-Rosenzweig, Cenlin He Funding acquisition: Cenlin He, Fei Chen, Timothy Schneider, Roy Rasmussen Investigation: Ronnie Abolafia-Rosenzweig, Cenlin He, Fei Chen Methodology: Ronnie Abolafia-Rosenzweig, Cenlin He, Fei Chen Project Administration: Fei Chen, Timothy Schneider, Roy Rasmussen

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# High Resolution Forecasting of Summer Drought in the Western United States

Ronnie Abolafia-Rosenzweig<sup>1</sup>, Cenlin He<sup>1</sup>, Fei Chen<sup>1</sup>, Kyoko Ikeda<sup>1</sup>, Timothy Schneider<sup>1</sup>, and Roy Rasmussen<sup>1</sup>

<sup>1</sup>National Center for Atmospheric Research, Boulder, CO, USA

Abstract Drought monitoring and forecasting systems are used in the United States (U.S.) to inform drought management decisions. Drought forecasting efforts have often been conducted and evaluated at coarse spatial resolutions (i.e., >10-km), which miss key local drought information at higher resolutions. Addressing the importance of forecasting drought at high resolutions, this study develops statistical models to evaluate 1- to 3-month lead time predictability of meteorological and agricultural summer drought across the western U.S. at a 4-km resolution. Our high-resolution drought predictions have statistically significant skill ( $p \le 0.05$ ) across 70%-100% of the western U.S., varying by evaluation metric and lead time. 1- to 3-month lead time drought forecasts accurately represent monitored summer drought spatial patterns during major drought events, the interannual variability of drought area from 1982 to 2020 (r = 0.84-0.93), and drought trends (r = 0.94-0.97). 71% of western U.S summer drought area interannual variability can be explained by cold-season (November-February) climate conditions alone allowing skillful 3-month lead time predictions. Pre-summer drought conditions (represented by drought indices) are the most important predictors for summer drought. Thus, the statistical models developed in this study heavily rely on the autocorrelation of chosen agricultural and meteorological drought indices which estimate land surface moisture memory. Indeed, prediction skill strongly correlates with persistence of drought conditions ( $r \ge 0.73$ ). This study is intended to support future development of operational drought early warning systems that inform drought management.

**Plain Language Summary** Droughts are complex and devastating natural disasters with severe implications in the western U.S. for energy production, agricultural yields, and wildfires, particularly during summer months. Drought monitoring and forecasting systems are used in the U.S. to inform drought management decisions. However, existing products and analyses are typically performed at coarse spatial resolutions (>10-km). Because summer drought is spatially variable at finer resolutions than current drought monitoring and forecasting systems, it is valuable to develop seasonal drought forecasting systems at higher spatial resolutions. In this study we present a novel statistical modeling framework to forecast summer drought conditions at a high resolution (4-km) at 1- to 3-month lead times across the western U.S. Our models provide skillful predictions of drought across most of the western U.S., but there is degraded skill in areas where drought tends to be less persistent. Future research can expand on our work by predicting drought at finer temporal resolutions while maintaining the high spatial resolution used in this study's methodology. This research is intended to support development of future operational drought early warning systems that provide high spatial resolution forecasts.

# 1. Introduction

Drought is a complex natural disaster with devastating economic and health consequences (Otkin et al., 2015). Since 1980, severe droughts have cost the United States (U.S.) at least \$249 billion, with an average cost of more than \$9.6 billion incurred during each drought event (NOAA NCEI, 2022). Drought consequences are particularly severe in the western U.S. (WUS). For instance, direct impacts from the widespread 2021 WUS drought cost approximately \$8.4 billion and resulted in hundreds of deaths (NOAA NCEI, 2022). This extreme drought year also enabled the third largest fire season (June–September) in the WUS since the start of the 1984 satellite record, with corresponding costs estimated at \$10.6 billion. WUS water and food security are also strongly dependent on summer drought, where summer drought events have contributed to anomalously low crop yields (Kogan, 1997; Zipper et al., 2016). Irrigation supports food production amidst drought (Zipper et al., 2016), which however has come at the cost of unsustainable over-exploitation of surface (Wheeler et al., 2022) and sub-surface (Scanlon et al., 2012) reservoirs. The WUS is in a historically significant megadrought, with 2000–2021 being



Rasmussen

Chen

Resources: Fei Chen, Kyoko Ikeda, Roy

Supervision: Cenlin He, Fei Chen,

Timothy Schneider, Roy Rasmussen Validation: Ronnie Abolafia-Rosenzweig

Visualization: Ronnie Abolafia-

Writing – original draft: Ronnie Abolafia-Rosenzweig, Cenlin He, Fei

Writing - review & editing: Ronnie

Abolafia-Rosenzweig, Cenlin He, Fei

Rosenzweig, Cenlin He

Chen, Kyoko Ikeda

the driest 22-year period since at least 800 (Williams et al., 2022). WUS drought frequency is projected to increase through 2050 with corresponding projections of increased forest-fire activity (Abatzoglou et al., 2021; Strzepek et al., 2010). Given the profound historical consequences, contemporary severity, and future projections of droughts in the WUS, it is essential to continually improve drought monitoring and forecasting systems that inform policy and management designed to mitigate the consequences of droughts. Therefore, this study is motivated to improve summer drought forecasting abilities in the WUS through predicting drought at a much higher spatial resolution (4-km) than existing seasonal drought forecasting frameworks (typically >10-km).

Several drought monitoring systems are used in the U.S. to inform drought response. The National Centers for Environmental Information (NCEI) and the Climate Prediction Center (CPC) of the National Oceanic and Atmospheric Administration (NOAA) publishes weekly Palmer Drought Severity Index (PDSI) maps for the conterminous U.S. (CONUS) by climate divisions (https://www.ncei.noaa.gov/access/monitoring/weekly-palmers/; https://www.cpc.ncep.noaa.gov/products/monitoring\_and\_data/drought.shtml). NCEI also publishes monthly drought maps referred to as the North American Drought Monitor (NADM) which presents drought conditions throughout the continent (https://www.ncdc.noaa.gov/temp-and-precip/drought/nadm/maps). NASA's North American Land Data Assimilation System (NLDAS) is used to create the NLDAS Drought Monitor which publishes weekly agricultural drought maps based on soil moisture simulations from the Mosaic, Noah, and Variable Infiltration Capacity (VIC) land surface models (LSMs) on a 1/8° (~12-km) spatial grid (https://ldas. gsfc.nasa.gov/nldas/drought-monitor; Xia et al., 2014). Perhaps the most popular means of monitoring drought in the U.S. is the U.S. Drought Monitor (USDM) which combines expert opinion with metrological and hydrological data to produce weekly maps of categorical drought severity for the U.S. ranging from abnormally dry (i.e., D0) to exceptional drought (i.e., D4) at the county level (Svoboda et al., 2002; https://droughtmonitor.unl. edu/). The West Wide Drought Tracker (WWDT) uses 4-km observationally based data to monitor drought conditions across the WUS dating back to 1895 (Abatzoglou et al., 2017). These operational monitoring systems are essential to inform drought management and related decision making; however, they do not explicitly provide information about future drought conditions which are essential to early drought warning and proactive planning (Fontaine et al., 2014).

Drought forecasts are generally made with dynamical (e.g., AghaKouchak, 2014; Shukla et al., 2014; Yoon et al., 2012) or statistical (e.g., Brust et al., 2021; Hao et al., 2016; Madadgar & Moradkhani, 2013, 2014; Park et al., 2016) model simulations of drought-related variables. Many of these approaches have been reviewed in recent literature (Fung et al., 2020; Hao, Yuan, et al., 2017; Prodhan et al., 2022). Dynamical approaches often rely on global or regional climate models based on physical processes of the atmosphere, ocean, cryosphere and land surface. The dynamical approach is the most advanced tool for drought forecasting. For instance, dynamical forecasts from the North American Multimodel Ensemble (NMME) (Kirtman et al., 2014) play an important role in probabilistic prediction of drought (Hao, Xia, et al., 2017). However, dynamical modeling methods are computationally expensive, are often limited by coarse resolution simulations that require bias correction and downscaling, and provide uncertain estimates of precipitation (AghaKouchak, 2014; Hao, Yuan, et al., 2017).

On the other hand, relationships between contemporary drought with future drought have been leveraged to inform statistical drought forecasts which are operationally useful but provide relatively little information about physical mechanisms of drought evolution. For instance, streamflow anomalies (used to quantify the severity of hydrologic drought), soil moisture percentiles (used to quantify the severity of agricultural drought) and the widely used PDSI (Palmer, 1965) (used to quantify the severity of meteorological drought) tend to have strong seasonal persistence (Dai, 2011; Lakshmi et al., 2004; Madadgar & Moradkhani, 2013, 2014; Szép et al., 2005), allowing past and present drought to be used as a predictor of future drought. Winter and spring snowpack, commonly referred to as a natural water tower for the western U.S., also plays a critical role in modulating summer drought and aridity conditions and provides additional predictive information for summer drought (Abolafia-Rosenzweig et al., 2022a; Huning & AghaKouchak, 2020; Van Ioon, 2015). Statistical approaches, such as regression or machine learning models, are based on empirical relationships depicted in historical climate and drought records and are useful due to their ease of implementation, relatively high computational efficiency, and established success in providing useful drought forecasts in the U.S. (Brust et al., 2021; Hao et al., 2016) and other regions around the globe (AghaKouchak, 2015; Hao et al., 2014; Mathivha et al., 2020; Rhee & Im, 2017). For instance, statistical models are capable of forecasting regions where USDM drought classifications are likely to intensify or improve (Lorenz et al., 2017; Otkin et al., 2013; https://www.drought.gov/) and USDM classifications up to 12 weeks in advance (Brust et al., 2021; Hao et al., 2016). Statistical models were also capable of forecasting the extreme 2012 U.S. drought several months in advance (AghaKouchak, 2014). Although these statistical drought forecasting frameworks can be valuable for informing drought management, these analyses have been conducted over coarse resolutions (>10-km) and thus provide limited information for forecasting the spatial heterogeneity of drought evolution at finer scales which is valuable for end-users (Samaniego et al., 2019).

Because summer WUS drought is spatially variable at finer resolutions than current forecasting systems with typical resolutions >10-km, and has close relationships with food security and wildfire in the WUS, it is imperative to develop seasonal drought forecasting systems at high spatial resolutions (e.g., <5-km; Samaniego et al., 2019). This study uses high resolution (4-km) observation-based (whenever available) spatially and temporally continuous hydroclimate data to develop statistical models for predicting summer meteorological and agricultural drought at a higher spatial resolution than previous U.S. drought forecasting analyses and existing products (AghaKouchak, 2014; Brust et al., 2021; Hao et al., 2016). Importantly, we quantify the predictability of drought at this higher spatial resolution which is akin to nowadays convection-permitting modeling efforts (Chen et al., 2020; Liu et al., 2017; Prein et al., 2015). In this study, we evaluate the ability of statistical models to provide summer drought forecasts across the WUS on a 4-km grid at 1-, 2-, and 3-month lead times. This analysis also assesses the relative importance for a suite of pre-summer drought predictors and provides insights relating the spatial heterogeneity of drought predictability to drought persistence characteristics. The results of this research are intended to inform development of drought forecasting systems that leverage high-resolution ( $\leq$ 4-km) hydrometeorological observations and simulations.

# 2. Study Domain and Methods

## 2.1. Study Domain

The WUS domain considered in this study is comprised of the five western regions from the National Integrated Drought Information System (NIDIS) drought early warning system (DEWS) (https://www.drought.gov/dews)— Pacific Northwest, California-Nevada, Missouri River Basin, Intermountain West, and Southern Plains regions that are north of 31°N latitude and west of 96°W longitude. This domain is spatially heterogenous, including a spectrum of arid to wet regions, deserts to forests, large irrigated agricultural areas, and snow-dominated mountains. The domain's hydrologic cycle is temporally variable, with distinct wet and dry seasons, particularly in the California-Nevada region. This domain is prone to crippling drought events, including the contemporary WUS megadrought (Williams et al., 2020, 2022). Droughts in the WUS can take many different forms including metrological, agricultural, hydrological, socioeconomic and snow drought (Huning & AghaKouchak, 2020; Littell et al., 2016; Livneh & Hoerling, 2016; Luo & Wood, 2007; Ryu et al., 2010; Westerling et al., 2006; Wlostowski et al., 2022). In this study we focus on meteorological and agricultural drought given their established relationships with wildfire, water security and food security in this domain as well as persistence of drought indices related to these categories (Abatzoglou & Kolden, 2013; Lakshmi et al., 2004; Littell et al., 2009; Lobell et al., 2013, 2014; Scanlon et al., 2012).

# 2.2. Seasonal Drought Forecasting Framework

The proposed drought forecasting framework (Figure 1) predicts seasonally averaged meteorological and agricultural drought indices (PDSI and soil moisture percentile, respectively) during summer months (June–August) at 1-, 2-, and 3-month lead times. PDSI and soil moisture percentiles are widely used drought indices in research applications and by the operational USDM because of their interpretable representation of land surface and meteorological drought conditions as spatiotemporally continuous fields (Abatzoglou & Kolden, 2013; Dai, 2011; T. W. Ford & Quiring, 2019; Kumar et al., 2014; Luo & Wood, 2007; Palmer, 1965; Shukla et al., 2011). PDSI is a commonly used meteorological drought index. It implicitly accounts for soil moisture conditions making it a less explicit representation of on-going meteorological conditions relative to other indices (e.g., SPI; T. Ford & Labosier, 2014), but results in relatively longer persistence making it more predictable by statistical models (Dai, 2011; Lakshmi et al., 2004; Szép et al., 2005). Figure S1 in Supporting Information S1 shows moderate correlations between summer PDSI and soil moisture percentiles across the western U.S (mean r = 0.63; interquartile range (IQR) = 0.57–0.74) indicating these indices are related but mostly independent.

The forecasting framework used herein adopts generalized additive models (GAMs; see Section 2.3 for details) using climate predictors (Table 2) averaged across winter and spring months of the same water year. The 1-month lead time forecasts use predictors averaged from November to April so predictions can be made by the end of







April. The 2-month lead time forecasts use predictors averaged from November-March so predictions can be made by the end of March. The 3-month lead time forecasts use predictors averaged from November to February so predictions can be made by the end of February. Preliminary tests were conducted to compare GAMs using predictors averaged beginning in November versus predictors averaged over the two most recent months available for the forecast (e.g., 1-month lead time forecasts using predictors averaged over March–April). This preliminary analysis concluded that predictions tend to be more accurate in the former case, using predictors starting in November. One reason for this is that accumulated drought conditions from winter through spring, rather than just spring, tend to provide a stronger land surface moisture memory that modulates summer drought conditions. This is supported by PDSI and soil moisture percentiles averaged over November–April tending to have higher correlations with summer (JJA) drought indices, relative to JJA comparisons with MA (March–April), FMA (February–April) and NDJ (November–January) averaged drought indices (Figure S2 in Supporting Information S1).

#### 2.3. Generalized Additive Models (GAMs)

All models in this study are trained or evaluated over a 39-year period (1982–2020) when climate and drought data are available (Table 2). We employ the widely used generalized additive models (GAMs) (Hastie & Tibshirani, 1986) using a Gaussian distribution to predict metrological and agricultural summer drought with pre-summer climate and drought conditions on 4-km grids across the WUS. GAMs are trained in the statistical software R (R Core Team, 2020) using the mgcv package (Wood, 2011). GAMs are logistic regression models, similar to normal linear regression models, but replace the linear form of  $\sum \beta_j X_j$  with a sum of nonlinear smooth functions  $\sum s_j(X_j)$ . Initial testing was performed using a more complex machine learning algorithm (Artificial Neural Networks; ANNs); however, ANNs were found to be too computationally expensive to comprehensively train across the domain (comprised of 2,57,216 4-km pixels), which is not suitable for potential operation applications. Thus, this study uses GAMs for prediction and analyses due to their success in previous drought and fire prediction research (Abolafia-Rosenzweig et al., 2022a, 2022b; Davenport et al., 2015) and their high computational efficiency which is essential for operational systems.

Climate predictors are pre-treated with a principal component analysis (PCA) to remove linear correlations among each predictor set (Dorman et al., 2013). For GAMs using less than five predictors, all PCs are used as predictors. For GAMs using five or more predictors, the first four PCs are used as predictors, which explain an average of 93% of the variance of all considered PCs. We limit the number of predictors to four to reduce model overfitting while allowing the large majority of climate predictor information to be explained by PC model inputs. GAMs used in this study can be written as:

$$g(E(\text{drought index})) = s(\text{PC}_1) + \dots + s(\text{PC}_n)$$
(1)

where  $g(\cdot)$  is a Gaussian link function, E(drought index) is the expected value of a drought index given the set of predictors,  $s(\cdot)$  is a nonlinear smooth function of climate predictors, PC<sub>i</sub> is the *i*th principal component for a set of climate predictors, and n is the number of climate predictors used in each GAM. Hence in this study, n never exceeds four. Drought indices predicted in this analysis are soil moisture percentiles (for agricultural drought) and PDSI (for meteorological drought). We generate an ensemble of GAMs by using unique combinations of predictors to ensure that results are insensitive to model-specific characteristics. Additionally, the ensemble of GAMs allows for probabilistic drought forecasting (AghaKouchak, 2014; Hao et al., 2014). Twenty best-performing ensemble members (evaluated based on all unique combinations of pre-summer predictors described in Table 2; totally 511 or 1,023 ensemble members tested for non-snowy and snowy pixels, respectively) are chosen based on a minimized Generalized Cross Validation (GCV) score. GCV is used for smoothness selection in the mgcv package where smoothing parameters are chosen to minimize prediction error. Ensembles chosen on the basis of GCV slightly outperforms ensembles chosen on the basis of the Akaike information criterion (AIC) (Table S1 in Supporting Information S1) so we exclusively consider ensembles chosen on the basis of GCV in this study. Each pixel is modeled independent of surrounding pixels, thus ensembles are trained and evaluated at each 4-km pixel. GAM simulations of soil moisture percentiles are post-processed to maintain the soil moisture percentile space, constrained between 0 and 1, using the following Equation 2:

soil moisture percentiles = 
$$\frac{SM_{rank} - min(SM_{rank})}{max(SM_{rank}) - min(SM_{rank})}$$
(2)

where SM<sub>rank</sub> is the original 39-value time series as ranks.

#### 2.4. Model Evaluation

Because GAMs are prone to overfitting, we exclusively evaluate model performance using out-of-bag comparisons (i.e., leave-one-out cross validations and retroactive forecasting) as done in previous evaluations of drought predictability (Bachmair et al., 2017; Davenport et al., 2015; Hao, Xia, et al., 2017). In leave-one-year-out cross validations, models are trained with data from all years except the target predicted year. Predictions are made for each year in the study period following this procedure resulting in a complete out-of-bag predicted summer drought time series. Retroactive forecasting mimics an operational forecasting system where models are trained with records from years prior to a target year. Retroactive forecasts in this analysis were made at 1-month lead times from 2010 to 2020 to support the leave-one-year-out cross validation. We assess the ability of GAMs to predict summer drought using three evaluation metrics: drought prediction accuracy (A), probability of detection (POD) and Pearson's correlation coefficient (r) as done in previous drought-related studies (Bhardwaj & Mishra, 2021; Fankhauser et al., 2022; T. W. Ford & Quiring, 2019; Hao & AghaKouchak, 2014; Zhu et al., 2016). Evaluation metrics compare GAM ensemble means with monitored drought. A represents the fraction of total predictions that are correct, ranging from 0 to 1, with 1 being a perfect score:

$$A = \frac{\text{hits} + \text{correct negatives}}{\text{total}}$$
(3)

POD quantifies the fraction of monitored droughts (Section 3) that are correctly predicted as drought by GAMs, ranging from 0 to 1, with 1 being a perfect score:

$$POD = \frac{hits}{hits + misses}$$
(4)

Hits represent summers when drought (D1–D4) was monitored and correctly predicted by GAMs. *Correct negatives* represent summers when abnormally dry (D0) to wet conditions are monitored and correctly predicted. *Total* 

# Table 1

United State Drought Monitor (USDM) Drought Classifications for PDSI and Soil Moisture Percentiles That Are Used in This Study (Svoboda et al., 2002)

Category	Description	Palmer drought severity index (PDSI)	Soil moisture model (percentiles)
D0	Abnormally dry	$-1 \ge PDSI > -2$	$30\% \ge SM$ percentile > 20%
D1	Moderate drought	$-2 \ge PDSI > -3$	$20\% \ge SM$ percentile > $10\%$
D2	Severe drought	$-3 \ge PDSI > -4$	$10\% \ge SM$ percentile > 5%
D3	Extreme drought	$-4 \ge PDSI > -5$	$5\% \ge SM$ percentile > $3\%$
D4	Exceptional drought	$-5 \ge PDSI$	$3\% \ge SM$ percentile

is the total number of summers evaluated (1982 through 2020 = 39 summers). *Misses* represent summers when drought is monitored but GAMs predict abnormally dry to wet conditions. The Pearson correlation (*r*) coefficient (Pearson & Henrici, 1896), ranging from -1 to 1, is widely used to quantify linear relationships between two variables, and thus serves as a well-known benchmark for performance:

$$r = \sum_{t=1}^{\text{total}} \frac{\left(M_t - \overline{M}\right) \left(P_t - \overline{P}\right)}{\sqrt{\sum_{t=1}^{\text{total}} \left(M_t - \overline{M}\right)^2 \left(P_t - \overline{P}\right)^2}}$$
(5)

where  $M_t$  and  $P_t$  represent monitored and GAM predicted drought indices at timestep t, respectively.  $\overline{M}$  and  $\overline{P}$  represent the time series mean of monitored and GAM predicted drought indices, respectively. For calculation of the A and POD metrics, droughts are considered as a binary event that occurs when drought indices fall into the USDM moderate to exceptional drought classification range (D1–D4) (Svoboda et al., 2002; Table 1). Statistical significance of A and POD metrics are computed from nonparametric bootstrap hypothesis tests, based on comparisons of random samples with monitored drought conditions at 10,000 randomly selected pixels.

Probabilistic drought forecasts can improve drought early warning systems (Hao et al., 2014). In this study, the forecasted drought probability is computed as the fraction of ensemble members that predict a drought event based on a specified threshold. We evaluate the ability of our ensemble modeling framework to provide useful

#### Table 2

```
Data Sets for Climate Predictors and Drought Indices
```

Variables	Data sources	Spatial resolution	Temporal resolution	Temporal record	Reference/location
Precipitation, temperature, vapor pressure deficit (VPD)	Parameter-elevation Relationships on Independent Slopes Model (PRISM)	4-km	Daily	1981—present	Daly et al. (2008, 2015) https://prism.oregonstate. edu/recent/
Palmer Drought Severity Index (PDSI)	Derived from PRISM precipitation and temperature	4-km	Monthly	1981—present	https://github.com/JonKing93/ pdsi/releases/tag/v1.0.0
Evapotranspiration (ET), potential ET (PET) and soil moisture	NCAR/USGS 4-km long-term CONUS hydroclimate reanalysis data set (CONUS404)	4-km	Hourly	1980–2021	Rasmussen et al. (2023)
Snow water equivalent (SWE)	Daily snow water equivalent and snow depth from assimilated in situ and modeled data over the Conterminous U.S., Version 1 (UA-SWE)	4-km	Daily	1981–2020	Broxton et al. (2019) Zeng et al. (2018) https://nsidc. org/data/nsidc-0719/ versions/1
Atlantic Multidecadal Oscillation Index (AMO)	NOAA Physical Sciences Laboratory AMO unsmoothed timeseries	North Atlantic (0°N -70°N)	Monthly	1948—present	Enfield et al. (2001) https:// psl.noaa.gov/data/ timeseries/AMO/
Pacific Decadal Oscillation (PDO)	NCEI PDO	North Pacific (N. of 20°N)	Monthly	1854—present	Mantua (1999) https:// www.ncdc.noaa.gov/ teleconnections/pdo/

Note. All 10 variables in this table are used as predictors.

probabilistic drought forecasts through comparisons of probabilistic drought maps from GAMs with monitored agricultural and meteorological drought maps for several significant summer drought events in 2007, 2011, 2012, and 2014. These evaluations consider drought thresholds of D1 and D3.

#### 2.5. Predictor Importance

Leave-one-column-out (LOCO) analyses are used to quantify relative predictors' importance at each pixel (Abolafia-Rosenzweig et al., 2022b; Kuhn-Régnier et al., 2021). The LOCO importance is measured by repeatedly training GAMs, each time without one particular predictor, and computing the ratio of a skill metric (Taylor Skill Score (*S*); Taylor, 2001) for the re-trained model to the original model including all of its original predictors. Ratios less than unity indicate degraded performance from removing a predictor, (i.e., reduced *S*), where *S* summarizes root mean square error (RMSE), ratio of variance and *r* as a single metric.

# 3. Data Used for Climate Predictors and Drought Indices

The climate predictors (Table 2) selected in this study are based on previous research that quantified and discussed relationships between these factors with drought in the U.S. (AghaKouchak, 2014; Brust et al., 2021; Hao et al., 2018; Lakshmi et al., 2004; McCabe et al., 2004; Otkin et al., 2018; Rajagopalan et al., 2000; Ryu et al., 2010). Data sources for these selected climate predictors and drought conditions come from long-term spatially and temporally continuous high-resolution data products, which are widely used and observation-based (whenever available) (Table 2). Specifically, precipitation, temperature and vapor pressure deficit (VPD) are obtained from the daily 4-km observation-based PRISM (Parameter-elevation Relationships on Independent Slopes Model) data set (Daly et al., 2008, 2015). PDSI is derived from PRISM precipitation and temperature, and snow water equivalent (SWE) is from the 4-km observation-based University of Arizona (UA) data set (Broxton et al., 2019; Zeng et al., 2018). Soil moisture, evapotranspiration (ET) and potential ET (PET) are from a new NCAR/USGS 4-km 40-year-plus (1980–2021) CONUS hydroclimate reanalysis data set (hereafter CONUS404; Rasmussen et al., 2023). Atlantic Multidecadal Oscillation (AMO) is published by NOAA's Physical Sciences Laboratory (Enfield et al., 2001), and Pacific Decadal Oscillation is published by NCEI (Mantua, 1999).

PDSI and root-zone (top 1 m) soil moisture percentiles are used to quantify meteorological and agricultural drought conditions, respectively. We use PRISM precipitation and temperature data to compute monitored PDSI because PRISM is an observationally based data set which is widely used and has been formally validated (Abatzoglou, 2013; Buban et al., 2020; Currier et al., 2017; Daly et al., 2008, 2015). PDSI is calculated using the pdsi MatLab package (King, 2022; https://github.com/JonKing93/pdsi/releases/tag/v1.0.0). Given the lack of long-term spatiotemporally continuous observational data products for soil moisture, evapotranspiration (ET) and potential ET (PET) at a high resolution ( $\leq$ 4-km) during the period of this study (1982–2020), we use the CONUS404 data product for these three variables. The CONUS404 data product is an extension of a widely used convection-permitting Weather Research & Forecasting (WRF) CONUS climate modeling product with improved model physics and configurations (Barlage et al., 2021; C. He et al., 2021; Liu et al., 2017). Many studies have shown that the convection-permitting WRF climate modeling product adequately captures key observed meteorological fields (e.g., precipitation, temperature, snowpack, and land surface states) (C. He et al., 2019; Ikeda et al., 2021; Liu et al., 2017; Scaff et al., 2020). In this study, we further evaluate the CONUS404 soil moisture, PET and ET data against in situ observations and satellite retrievals (see below for details) before using them in the GAMs model training and forecasting.

Figure S3 in Supporting Information S1 and Figure 2 present comparisons of CONUS404 soil moisture percentiles with 509 in situ monitoring stations across the WUS and operationally used drought monitoring products, respectively. CONUS404 drought predictability is benchmarked against widely used LSMs from Phase 2 of the North American Land Data Assimilation System (NLDAS-2; Xia, Ek, et al., 2012, Xia, Mitchell, et al., 2012) that is also currently used for operationally monitoring U.S. drought. In situ soil moisture data from 1997 to 2019 and nearest-neighbor matched NLDAS-2 Noah soil moisture data are from T. W. Ford and Quiring (2019). CONUS404 has similar agreement with in situ observations relative to the NLDAS-2 Noah LSM, with a mean prediction accuracy (*A*) of 0.65–0.75 and a mean POD of 0.8–0.9 (Figure S3 in Supporting Information S1). Furthermore, CONUS404 monitored agricultural drought conditions during four major summer drought events in the study record (2007, 2011, 2012, and 2014) well capture operationally monitored NLDAS-2 ensemble mean





**Figure 2.** Monitored drought conditions during four major summer drought events in July of 2007, 2011, 2012. 4-km drought monitoring products for meteorological drought (Palmer Drought Severity Index: PDSI) and agricultural drought (soil moisture percentiles) used in this study are outlined in dashed green rectangles. Drought monitoring products used in this study (PRISM and CONUS404) show mostly consistent spatial distributions of drought conditions during these drought events relative to operationally used drought monitoring products that use corresponding drought indices as well as the integrated drought index represented by the USDM. NLDAS-2 agricultural drought is based on the multi-model (VIC, Noah and Mosaic) ensemble mean. CPC and USDM weekly drought maps are from the last week of July from respective years.

drought conditions (Figure 2). Similarly, PRISM-derived PDSI favorably compares to operationally monitored PDSI conditions from CPC and NCEI archives (https://www.cpc.ncep.noaa.gov/products/monitoring\_and\_data/ drought.shtml; https://www.ncdc.noaa.gov/temp-and-precip/drought/historical-palmers/maps). CONUS404 agricultural drought index and PRISM PDSI also compare favorably with the integrated drought maps and characteristics from USDM during these summer drought events (Figure 2). Thus, given the adequate accuracy of CONUS404 in simulating soil moisture percentiles and agricultural drought and its high spatial resolution, we consider the CONUS404 soil moisture data a valuable agricultural drought monitoring product to use in this analysis. The wide operational use and previous evaluations of PRISM combined with Figure 2 further support the use of PRISM PDSI as a meteorological drought monitoring product.

We also evaluate CONUS404 ET and PET, which are exclusively used as predictors. CONUS404 ET is obtained directly from the CONUS404 outputs and PET is derived from CONUS404 outputs following the Penman-Monteith equation (Technical Committee on Standardization of Reference Evapotranspiration, 2005; Yodder et al., 2005). CONUS404 ET and PET are compared with the widely used Moderate Resolution Imaging Spectroradiometer (MODIS) MOD16A2 data set (Mu et al., 2013) across the WUS during 2001–2020. CONUS404 ET and PET agree with MOD16A2 ET and PET across the WUS (median r = 0.75 and 0.96, respectively; Figure S4 in Supporting Information S1). Areas of low correlation between CONUS404 and MOD16A2 ET correspond with arid areas characterized by heavily water-limited ET (Figure S4 in Supporting Information S1). Overall, we consider CONUS404 ET and PET reasonable estimates over most of the western CONUS that can provide valuable predictive information in GAMs.

As a pre-processing step to statistical modeling, all variables are re-mapped to the CONUS404 4-km grid using bilinear interpolation. Bilinear interpolation is adequate because all data have the same native spatial resolution (4-km), excluding the Atlantic Multidecadal Oscillation (AMO) and the Pacific Decadal Oscillation (PDO) indices which are not spatially distributed, and thus are considered uniform across the domain. Predictors and drought conditions are first averaged to a monthly time scale, then predictors are temporally averaged over pre-summer months and monitored drought indices are averaged over summer months (Figure 1). SWE is only used as a predictor of snowy pixels that may be affected by snow processes: pixels that recorded less than 10 years of 0 mm pre-summer SWE and at least 1 year of at least 0.1 mm of mean pre-summer SWE. This results in 83% of pixels considering SWE as a predictor and 17% pixels not considering SWE as a predictor in the study domain.

# 4. Results

# 4.1. Evaluating GAM 1-Month Lead Time Predictions

GAMs provide skillful 1-month lead time predictions of meteorological and agricultural summer drought across most of the WUS at a 4-km resolution while maintaining accurate representations of the total WUS area effected by various drought severities and long-term trends (Figures 3–5). In leave-one-year-out cross validations A, POD and r for meteorological drought predictions are statistically significant ( $p \le 0.05$ ) across 93%, 85%, and 100% of the domain with median values of 0.90, 0.67, and 0.86, respectively (Figures 3a, 3c, and 3e). For agricultural drought predictions, A, POD and r are statistically significant ( $p \le 0.05$ ) across 77%, 94%, and 100% of the domain with median values of 0.85, 0.63, and 0.78, respectively (Figures 3b, 3d, and 3f). Retroactive forecasting also provides statistically significant skill of drought predictions across the large majority of the western U.S. (95% and 89% of the domain for meteorological and agricultural drought, respectively) where median correlations for predicted PDSI and soil moisture percentiles are 0.84 and 0.77, respectively (Figure S5 in Supporting Information S1). Median and interquartile ranges (IQR) of evaluation metrics for drought predictions are summarized in Table 3. 39-year linear trends of predicted drought index time series closely match monitored trends with high correlations (r = 0.94-0.97) and acceptable biases (-18%-3%) across all pixels in the WUS (Figure 4). Spatially averaged monitored and predicted drought indices each show negative trends for PDSI (-0.0214/year and -0.0207/year) and soil moisture percentiles (-0.0028/year and -0.0024/year, respectively), indicating that GAMs accurately captured the direction of broadscale drought intensification from 1982 to 2020.

Predicted drought area (D1 or more severe) has high correlations (0.90–0.98;  $p \le 0.01$ ) and low to moderate biases (-23%–0%) with monitored agricultural and meteorological drought area, respectively from leave-one-year-out cross validations (Figure 5) and retroactive forecasting analyses (Figure S6 in Supporting Information S1). The leave-one-year-out correlation is degraded to 0.82 for meteorological drought area when the threshold is increased to extreme drought (D3), and further degraded to 0.68 when only considering exceptional drought (D4). The correlation between predicted and monitored agricultural drought is mostly insensitive to drought severity (ranging from 0.89 to 0.92). One reason for this insensitivity for agricultural drought area, compared to the relatively





**Figure 3.** Evaluation metrics for GAM simulated metrological (left) and agricultural (right) drought with 1-month lead times in leave-one-year-out cross validations. (a), (b) Spatial maps of the fraction of correct predictions (*A* from Equation 3) (c), (d) spatial maps of probability of drought detection (POD from Equation 4). (e), (f) Spatial maps of correlation (*r* form Equation 5). The lower bound of color bars is the p = 0.05 significance level.

higher sensitivity for meteorological drought area, is that soil moisture percentile predictions are post-processed to maintain the known percentile space (Equation 2), whereas predicted PDSI is not post-processed because PDSI has temporally and spatially variable ranges. Hence, uncertainties in predicted drought area can be ameliorated by understanding and accounting for the observed drought index space. As shown in Text S1 and Figure S13 of the Supporting Information S1, converting monitored and predicted PDSI to the percentile space reduces GAM-predicted PDSI errors, particularly during abnormally wet or dry (i.e., drought) times and places.

Predicted drought has similar spatial distributions relative to monitored meteorological and agricultural drought during noteworthy drought events in both leave-one-year-out (Figure 6) and retroactive forecasting (Figure S7 in Supporting Information S1) cross validations. Hence, accurate drought area predictions from GAMs shown





Figure 4. (a) Monitored and (b) 1-month lead time predicted linear trends in PDSI. (d) Monitored and (e) 1-month lead time predicted linear trends in soil moisture percentiles form 1982–2020. Biases in predicted trends for (c) PDSI and (f) soil moisture percentiles. Correlations between (a), (b) and (d), (e) are 0.97 and 0.94, respectively.



**Figure 5.** Summer drought areal coverage (as fraction of total WUS domain) from drought monitoring data sources and GAM 1-month lead time predictions in leave-one-year-out cross validations. (a) Monitored meteorological drought from PRISM PDSI. (b) GAM ensemble mean of predicted meteorological drought. (c) Monitored agricultural drought from CONUS404 soil moisture percentiles. (d) GAM ensemble mean of predicted agricultural drought. (e) Comparison of monitored and GAM simulated meteorological drought area for various drought thresholds with corresponding r reported for each. (f) Comparison of monitored and GAM simulated agricultural drought area for various drought thresholds with corresponding r reported for each.

# Table 3

Evaluation Metrics Comparing GAM Ensemble Mean Leave-One-Year-Out Predictions With Monitored Meteorological and Agricultural Drought

	Correct predictions (A)	Probability of detection (POD)	Correlation ( <i>r</i> )	Percent of pixels with significant A (%)	Percent of pixels with significant POD (%)	Percent of pixels with significant r (%)			
Meteorological drought evaluation metrics									
1-month lead time	0.90 [0.85-0.92]	0.67 [0.54–0.79]	0.86 [0.80-0.90]	93	85	100			
2-month lead time	0.87 [0.85-0.92]	0.66 [0.50-0.75]	0.84 [0.78–0.88]	90	82	100			
3-month lead time	0.87 [0.85-0.92]	0.64 [0.50-0.75]	0.83 [0.78–0.87]	89	81	100			
Agricultural drought evaluation metrics									
1-month lead time	0.85 [0.85-0.90]	0.63 [0.63-0.75]	0.78 [0.70-0.86]	77	94	100			
2-month lead time	0.85 [0.79-0.90]	0.63 [0.50-0.75]	0.75 [0.67–0.84]	70	91	100			
3-month lead time	0.85 [0.79–0.90]	0.63 [0.50-0.75]	0.75 [0.67–0.83]	70	91	100			
Meteorological drough 1-month lead time 2-month lead time 3-month lead time Agricultural drought et 1-month lead time 2-month lead time 3-month lead time	tt evaluation metric: 0.90 [0.85–0.92] 0.87 [0.85–0.92] 0.87 [0.85–0.92] valuation metrics 0.85 [0.85–0.90] 0.85 [0.79–0.90] 0.85 [0.79–0.90]	s 0.67 [0.54–0.79] 0.66 [0.50–0.75] 0.64 [0.50–0.75] 0.63 [0.63–0.75] 0.63 [0.50–0.75] 0.63 [0.50–0.75]	0.86 [0.80–0.90] 0.84 [0.78–0.88] 0.83 [0.78–0.87] 0.78 [0.70–0.86] 0.75 [0.67–0.84] 0.75 [0.67–0.83]	93 90 89 77 70 70	85 82 81 94 91 91	100 100 100 100 100 100			

Note. Predictions are made with 1-, 2-, and 3-month lead times. Metrics are reported as: median [inter quartile range].

in Figure 5 and Figure S6 in Supporting Information S1 correspond with accurate spatial distributions during noteworthy drought events. The higher accuracy in the leave-one-year-out 2011 summer drought prediction (Figure 6), relative to the retroactive forecast (Figure S7 in Supporting Information S1), emphasizes that using a relatively longer statistical model training period can yield more skillful drought predictions, and thus emphasizes the importance of maintaining historical drought records.



Figure 6. Monitored meteorological and agricultural drought classifications from PRISM and CONUS404 in top rows of (a) and (b), respectively. Predicted meteorological and agricultural drought classifications from GAMs in leave-one-year-out cross validations in respective lower rows of (a) and (b). Predictions are made with a 1-month lead time.

During the same four drought events shown in Figure 6 (2007, 2011, 2012, and 2014 summers), Figure 7 compares probabilistic drought predictions (using a D1 threshold; see Section 2.2) with monitored drought, revealing that GAM ensembles predict relatively high probabilities of drought over areas and times where drought is monitored. Specifically, the median and IQR of predicted meteorological drought probability for pixels where drought was monitored during these events is 0.80 and 0.55–0.95, respectively; whereas the median and IQR for pixels with no meteorological drought monitored is 0.0 and 0.0–0.05, respectively. Similarly, the median and IQR of predicted agricultural drought probability for pixels where drought was monitored during these events is significantly higher (0.35 and 0.15–0.60, respectively; p < 0.01) than pixels where no drought is monitored (0.0 and 0.0–0.05). This result is similar when drought is considered as D3 or more severe (Figure S8 in Supporting Information S1). Thus, GAMs are capable of accurately forecasting spatial distributions of major drought events (including exceptional drought) at a 1-month lead time.

There is substantial spatial heterogeneity in high resolution drought prediction skill (Figure 3). Summer drought predictions tend to be relatively inaccurate along the northern west coast from Northern California through Washington, the Sierra Nevada, Northern Idaho, Northern Texas and Oklahoma. Some regions where summer drought is particularly predictable are Southern California and the Central Valley, agricultural plains of Southern Idaho along the Snake River, Nebraskan agricultural plains along the Platte River, and North Dakota and South Dakota along the Missouri River. One factor that affects drought predictability is the persistence of drought, where drought predictability is expected to decrease in areas with relatively weak drought persistence (Hao et al., 2018). We quantify drought persistence as the correlation between mean pre-summer and mean summer drought indices (Figures 8a and 8b). Drought persistence maps show similar features as GAM skill, particularly for r (Figures 3e and 3f), where regions with low drought persistence generally have low prediction skill. Indeed, when evaluation metrics are binned by drought persistence, there is a tendency for areas with greater drought persistence to correspond with better skill scores (Figures 8d–8f). For instance, there are high correlations between drought persistence (i.e., Figures 8a and 8b) and GAM leave-one-year-out r (i.e., Figures 3e and 3f), with correlations of 0.82 and 0.73 for PDSI and soil moisture percentiles, respectively (p < 0.01). Many factors modulate land surface moisture memory (i.e., drought persistence) (Koster & Suarez, 2001; Mahanama & Koster, 2003), with a noticeable influence from precipitation (Rahman et al., 2015) where drier areas tend to have greater persistence and more skill (Figure 8). However, there are many exceptions to this statement, as indicated by only moderate anti-correlations between mean summer precipitation with drought persistence: -0.40 and -0.39 (p < 0.01) for soil moisture percentiles and PDSI, respectively. One such exception is that drought tends to be relatively persistent and predictable along surface water channels in relatively rainy regions (e.g., Nebraska) (Figures 3f and 8b), indicating an important role of land surface characteristics in modulating drought persistence and predictability as well. Overall, drought predictability tends to be higher where drought is more persistent, which has significant overlap with areas that have relatively dry summers.

#### 4.2. Relative Importance of Predictors

Pre-summer PDSI is the most important predictor for summer PDSI, and pre-summer soil moisture percentile is the most important predictor for summer soil moisture percentile (Figure 9), as expected from the key role of drought persistence in controlling prediction skill discussed in Section 4.1. Thus, the GAMs used in this study heavily rely on the autocorrelation attribute of drought indices. Specifically, pre-summer PDSI is the most important predictor for summer soil moisture percentile over 48% of the WUS (Figures 9c and 9g). Pre-summer PDSI is also an important predictor of summer soil moisture percentile over 48% of the WUS (Figures 9c and 9g). Pre-summer PDSI is also an important predictor of summer soil moisture percentile, showing the greatest importance over 16% of the WUS. Given the dominant importance of the pre-summer drought indices, we also present the second most important predictors for GAMs over the domain (Figures 9b, 9d, 9f, and 9h). This reveals that meteorological drought predictability in the WUS also relies heavily on pre-summer VPD, temperature, soil moisture percentiles, PET, ET and precipitation, which vary by regions. These are the second most important predictor for summer PDSI over at least 11% of the WUS, whereas the rest of predictors are the second most important predictor over less than 6.5% of the WUS. Nine of the 10 predictors are the second most important predictor over less than 6.5% of the WUS, whereas SWE is the second most important predictor over only 3% of the WUS.

However, SWE is an important predictor for PDSI in North Dakota (Figure 9b) and soil moisture percentiles in the Sierra Nevada (Figure 9c), consistent with previous research showing connections between pre-summer snow conditions with summer hydrology and drought in these areas (Dierauer et al., 2019; M. He et al., 2016; Quiring





**Figure 7.** Monitored meteorological and agricultural binary drought (D1–D4) from PRISM and CONUS404 in top rows of (a) and (b), respectively. Probabilistic predictions of meteorological and agricultural drought probability from GAMs in leave-one-year-out cross validations in respective second rows. Boxplots of predicted (c) meteorological and (d) agricultural drought probability grouped by pixels that are monitored to be in drought and no drought. Predictions are made with a 1-month lead time. For boxplots, central marks indicate the median, and the bottom and top edges of boxes indicate the 25th and 75th percentiles, respectively. Whisker lengths are equivalent to the interquartile range. Outliers are plotted as black dots.

& Kluver, 2009; Shin et al., 2020). The importance of SWE as a predictor of agricultural drought in the Sierra Nevada is also supported by Figure S2 in Supporting Information S1 which reveals soil moisture percentiles averaged over early winter months (November–January) have higher correlations with summer (JJA) soil moisture percentiles, relative to soil moisture percentiles averaged over late-winter and spring months (February–April) in this area.





**Figure 8.** Summer drought tends to be more predictable over areas where drought indices are more persistent and drought tends to be more persistent over areas with relatively low summer precipitation. (a) Correlation between pre-summer (November–April mean) PDSI and summer (June–August) PDSI. (b) Same as (a), but for soil moisture percentiles instead of PDSI. (c) Mean summer precipitation. (d) The *A* metric from leave-one-year-out cross validations for GAMs predicting PDSI (top) and soil moisture percentiles (bottom) uniformly binned by persistence of respective drought indices, where persistence is shown spatially in (a) and (b), respectively. (e) Same as (d) but for *r*. (g) Persistence of PDSI (top) and soil moisture percentiles (bottom) uniformly binned by mean summer precipitation. (h) *r* from leave-one-year-out cross validations for GAMs predicting PDSI (top) and soil moisture percentiles (bottom) uniformly binned by mean summer precipitation. (h) *r* from leave-one-year-out cross validations for GAMs predicting PDSI (top) and soil moisture percentiles (bottom) binned by mean summer precipitation. (h) *r* from leave-one-year-out cross validations for GAMs predicting PDSI (top) and soil moisture percentiles (bottom) binned by mean summer precipitation. (h) *r* from leave-one-year-out cross validations for GAMs predicting PDSI (top) and soil moisture percentiles (bottom) binned by mean summer precipitation. Persistence bins are divided into equal-length terciles such that: low persistence (<0.44), mid. persistence (0.44–0.73), high persistence (>0.73) for ag. drought and low persistence (<0.80), mid. persistence (0.80–0.86), high persistence (>0.86) for met. drought. For boxplots, central marks indicate the median, and the bottom and top edges of boxes indicate the 25th and 75th percentiles, respectively. Whisker lengths are equivalent to the interquartile range. Outliers are plotted as red dots.

#### 4.3. Lead Time Sensitivity: Evaluating 2- and 3-Month Lead Time Predictions

2- and 3-month lead time predictions of meteorological and agricultural summer drought are skillful across most of the WUS at a 4-km resolution while maintaining accurate representations of long-term trends and the total WUS area affected by various drought classifications (Table 3; Figures 10–13; Figures S9 and 10 in Supporting Information S1). For instance, 2- and 3-month lead time drought predictions have significant skill ( $p \le 0.05$ ) across 81%-100% and 70%-100% of the WUS for meteorological and agricultural drought, respectively (Figures 10 and 11). *A*, POD and *r* from 1-, 2-, and 3-month lead time predictions are similar (Table 3). 2- and 3-month lead time predictions of meteorological drought provide median *A*, POD, and *r* values of at least 0.87, 0.64, and 0.83, respectively. 2- and 3-month lead time predictions of agricultural drought provide median *A*, POD, and *r* values of at least 0.85, 0.63, and 0.75, respectively. 2- and 3-month lead time predictions maintain the spatial variability of monitored summer agricultural and meteorological drought trends (r = 0.94-0.96; Figures S9 and S10 in Supporting Information S1). GAM predicted drought area (D1 or more severe) has high correlations (0.84–0.89;  $p \le 0.01$ ) and low to moderate biases (0%–26%) with monitored agricultural and meteorological





Figure 9. Relative importance of predictors for PDSI and soil moisture percentiles based on leave-one-column-out analyses from 1-month lead time models. (a) Map of the most important predictor for PDSI. (b) Map of the second most important predictor for PDSI. (c) Map of the most important predictor for soil moisture percentiles. (d) Map of the second most important predictor for soil moisture percentiles. (e) Number of pixels each predictor of PDSI is the most important. (f) Number of pixels each predictor of soil moisture percentile is the second most important. (g) Number of pixels each predictor of soil moisture percentile is the second most important.

drought area (Figure 12). This correlation is degraded to 0.75 for meteorological drought area when the threshold is increased to extreme drought (D3), and further degraded to 0.63–0.65 when only considering exceptional drought (D4). The correlation between predicted and monitored agricultural drought is mostly insensitive to drought classification (ranging from 0.83 to 0.86). The high correlation between 3-month lead time drought area predictions (using winter predictors) with monitored summer drought area indicates the majority of the summer drought area's interannual variability can be explained by winter conditions alone. Predicted summer drought from 2- and 3-month lead time simulations show similar spatial distributions relative to monitored meteorological and agricultural drought during the 2007, 2011, 2012, and 2014 significant summer drought events (Figure 13). Thus, our GAM forecasting system can provide useful 2- and 3-month lead time predictions of summer drought at a high resolution.

The relative importance of predictors from 1-month lead time predictions are similar for models providing 2- and 3-month lead time predictions (Figures S11 and S12 in Supporting Information S1). For 2- and 3-month lead time predictions, pre-summer PDSI is the most important predictor for summer PDSI over 97% of the WUS



**Figure 10.** Evaluation metrics for GAM simulated metrological (left) and agricultural (right) drought with 2-month lead times in leave-one-year-out cross validations. (a), (b) Spatial maps of the fraction of correct predictions (*A* from Equation 3) (c), (d) spatial maps of probability of drought detection (POD from Equation 4). (e), (f) Spatial maps of correlation (*r* form Equation 5). The lower bound of color bars is the p = 0.05 significance level.

(Figures S11e and S12e in Supporting Information S1), and pre-summer soil moisture percentile is the most important predictor for summer soil moisture percentile over 52%–54% of the WUS (Figures S11g and S12g in Supporting Information S1), because of the dominant role of drought persistence. Pre-summer PDSI is also an important predictor of summer soil moisture at 2- and 3-month lead times, showing the greatest importance over 21% and 22% of the WUS, respectively. Meteorological drought predictability in the WUS also relies heavily on pre-summer soil moisture percentiles, temperature, VPD, PET, ET, and precipitation (Figures S11b, S11f, S12b, and S12f in Supporting Information S1). Pre-summer PDSI is most frequently (17%–18%) the second most important over 8%–12% of the WUS (Figures S11d, S11h, S12d, and S12h in Supporting Information S1).



# Water Resources Research



**Figure 11.** Evaluation metrics for GAM simulated metrological (left) and agricultural (right) drought with 3-month lead times in leave-one-year-out cross validations. (a), (b) Spatial maps of the fraction of correct predictions (*A* from Equation 3) (c), (d) spatial maps of probability of drought detection (POD from Equation 4). (e), (f) Spatial maps of correlation (*r* form Equation 5). The lower bound of color bars is the p = 0.05 significance level.

# 5. Discussion

Stakeholders in agriculture, energy, wildfire, and water resources sectors require accurate and detailed local drought seasonal forecasts so important proactive decision-making can be made based on the best available data (Abatzoglou et al., 2017). For instance, drought data informs crop water requirements, and in turn surface water and groundwater demands and water allocations for irrigation organizations that play a critical role in water distribution and groundwater management decision making, and long-range weather forecasts have been identified among the most commonly used data sources for long-range planning and management by irrigation organizations (Wallander et al., 2022). Wildland fire managers have an increasing need for drought data to be incorporated





**Figure 12.** Summer areal drought coverage (as fraction of total WUS domain) from drought monitoring data sources and GAM predictions in leave-one-year-out cross validations. (a), (b) Comparison of monitored and GAM 2-month lead time predicted meteorological and agricultural drought area for various drought thresholds with corresponding r reported for each. (c), (d) Comparison of monitored and GAM 3-month lead time predicted meteorological and agricultural drought area for various drought thresholds with corresponding r reported for each.

to fire-related forecasts to improve firefighter safety and response, public health safety, and long-term fuel treatment strategies. In response, improving S2S forecasting of drought and fire hazard is a priority of NOAA's Weather Program Office. There are multiple validated observational and model-based hydrometeolrogical data sets at higher resolutions (e.g.,  $\leq$ 4-km) than drought forecasting analyses in the WUS (Daly et al., 2008, 2015; Liu et al., 2017), motivating the research conducted in this analysis which leverages data sets with a 4-km spatial resolution to predict drought at S2S scales while maintaining the native spatial resolution of these data.

This study finds widely used meteorological and agricultural drought indices are predictable at a 4-km resolution at S2S time-scales across most of the WUS using a relatively simple statistical modeling methodology. However, drought predictability in this study is heavily degraded in areas where drought has low persistence. Future analyses may attempt to overcome this shortcoming by using spatially variable pre-summer temporal averaging to maximize the correlation with pre-summer and summer drought (i.e., Figure S2 in Supporting Information S1). Furthermore, future analyses may consider using other predictors that were not considered in this study, such as the Oceanic Niño Index which tracks the ocean part of the El Niño-southern Oscillation climate pattern which has important relationships with drought in the WUS (Mo & Schemm, 2008), more detailed snow indices (e.g., peak SWE, ablation rate, day of snow disappearance, snow drought, etc.) (M. He et al., 2016; Huning & AghaKouchak, 2020), S2S forecasts of North American Monsoon rainfall (Prein et al., 2022), and other information from long-range dynamical modeling forecasts using combined dynamical and statistical approaches (e.g., Yan et al., 2017). Future analyses that use statistical models to forecast drought should also explicitly account for the ranges of drought indices (as discussed in Section 4.1 and Text S1 in Supporting Information S1). Another limitation is that although the WUS drought predictions made in this study are at higher-resolutions than previous research, there is still a great deal of spatial





Figure 13. Monitored meteorological and agricultural drought classifications from PRISM and CONUS404 in rows one and four, respectively. 2-month lead time predicted meteorological and agricultural drought classifications from GAMs in leave-one-year-out cross validations in rows two and five, respectively. 3-month lead time predicted meteorological and agricultural drought classifications from GAMs in leave-one-year-out cross validations in rows three and six, respectively.

heterogeneity of drought at even higher resolutions (e.g., sub-kilometer scales) (Vergopolan et al., 2022). The methodology used in this analysis may be applicable at the sub-kilometer scale as more spatially and temporally continuous data sets become available at higher resolutions (e.g., Anderson et al., 2012; Burek et al., 2020; Vergopolan et al., 2021).

# 6. Conclusions

In this study we evaluated the ability of computationally efficient statistical models to forecast summer drought across the western U.S. from 1982 to 2020 at a much higher spatial resolution (4-km) than previous drought forecasting efforts. Summer drought predictors are pre-summer drought and climate conditions from long-term spatially and temporally continuous 4-km observation-based (whenever available) or reanalysis data sets. Our statistical forecasting models provide skillful predictions of meteorological and agricultural drought across the



western U.S. at 1- to 3-month lead times, and maintain accurate spatial and temporal variability of summer drought area. Our drought predictions provide statistically significant ( $p \le 0.05$ ) skill scores across 70%–100% of the domain, while explaining at least 71% of the monitored interannual variability of drought area for D1 or more severe events (USDM drought categorization). Our predictions provide similar drought spatial distributions relative to monitored drought conditions during four representative major summer drought events in 2007, 2011, 2012, and 2014. Drought predictability is spatially variable, with drought tending to be more predictable in drier areas where drought is more persistent. As expected, pre-summer PDSI is the most important predictor for summer PDSI (representing meteorological drought condition), and pre-summer soil moisture percentile is the most important predictor for summer soil moisture percentile (representing agricultural drought condition), emphasizing that the statistical models used in this study rely heavily on the autocorrelation attribute of drought conditions, consistent with previous studies. This work presents a new capability to predict seasonal drought conditions at a high spatial resolution across the western U.S. Future research should expand on this study by evaluating drought forecasting at higher temporal resolutions (e.g., monthly or weekly instead of seasonal scale) while maintaining a high spatial resolution. This study is intended to support the development of future operational drought early warning systems that provide high resolution drought forecasts.

# **Data Availability Statement**

The data that support the findings of this study are openly available: https://data.mendeley.com/datasets/gw7c3y-jhyp/2 (Abolafia-Rosenzweig et al., 2022c). The CONUS 404 data are archived and accessible on the United States Geological Survey Black Pearl Tape system and on the NCAR super-computer Campaign storage system.

# References

- Abatzoglou, J. T. (2013). Development of gridded surface meteorological data for ecological applications and modelling. *International Journal of Climatology*, 33(1), 121–131. https://doi.org/10.1002/joc.3413
- Abatzoglou, J. T., Battisti, D. S., Williams, A. P., Hansen, W. D., Harvey, B. J., & Kolden, C. A. (2021). Projected increases in western US forest fire despite growing fuel constraints. *Communications Earth & Environment*, 2(1), 227. https://doi.org/10.1038/s43247-021-00299-0
- Abatzoglou, J. T., & Kolden, C. A. (2013). Relationships between climate and macroscale area burned in the western United States. *International Journal of Wildland Fire*, 22(7), 1003. https://doi.org/10.1071/WF13019
- Abatzoglou, J. T., McEvoy, D. J., & Redmond, K. T. (2017). The west wide drought tracker: Drought monitoring at fine spatial scales. Bulletin of the American Meteorological Society, 98(9), 1815–1820. https://doi.org/10.1175/BAMS-D-16-0193.1
- Abolafia-Rosenzweig, R., He, C., & Chen, F. (2022a). For western wildfires, the immediate past is prologue. *Eos*, 103. https://doi.org/10.1029/2022EO220319
- Abolafia-Rosenzweig, R., He, C., & Chen, F. (2022b). Winter and spring climate explains a large portion of interannual variability and trend in western U.S. summer fire burned area. *Environmental Research Letters*, *17*(5), 054030. https://doi.org/10.1088/1748-9326/ac6886

Abolafia-Rosenzweig, R., He, C., Chen, F., Ikeda, K., Schneider, T., & Rasmussen, R. (2022c). High resolution ensemble forecasting of summer drought in the western United States with statistical models. *Mendeley Data*. V2. https://doi.org/10.17632/gw7c3yjhyp.2

- AghaKouchak, A. (2014). A baseline probabilistic drought forecasting framework using standardized soil moisture index: Application to the 2012 United States drought. *Hydrology and Earth System Sciences*, 18(7), 2485–2492. https://doi.org/10.5194/hess-18-2485-2014
- AghaKouchak, A. (2015). A multivariate approach for persistence-based drought prediction: Application to the 2010–2011 East Africa drought. *Journal of Hydrology*, 526, 127–135. https://doi.org/10.1016/j.jhydrol.2014.09.063
- Anderson, M. C., Allen, R. G., Morse, A., & Kustas, W. P. (2012). Use of Landsat thermal imagery in monitoring evapotranspiration and managing water resources. *Remote Sensing of Environment*, 122, 50–65. https://doi.org/10.1016/j.rse.2011.08.025
- Bachmair, S., Svensson, C., Prosdocimi, I., Hannaford, J., & Stahl, K. (2017). Developing drought impact functions for drought risk management. Natural Hazards and Earth System Sciences, 17(11), 1947–1960. https://doi.org/10.5194/nhess-17-1947-2017
- Barlage, M., Chen, F., Rasmussen, R., Zhang, Z., & Miguez-Macho, G. (2021). The importance of scale-dependent groundwater processes in land-atmosphere interactions over the central United States. *Geophysical Research Letters*, 48(5), e2020GL092171. https://doi. org/10.1029/2020GL092171

Bhardwaj, K., & Mishra, V. (2021). Drought detection and declaration in India. Water Security, 14, 100104. https://doi.org/10.1016/j.wasec.2021.100104

- Broxton, P., Zeng, X., & Dawson, N. (2019). Daily 4 km gridded SWE and snow depth from assimilated in-situ and modeled data over the conterminous US, version 1. [Indicate subset used]. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi. org/10.5067/0GGPB220EX6A
- Brust, C., Kimball, J. S., Maneta, M. P., Jencso, K., & Reichle, R. H. (2021). DroughtCast: A machine learning forecast of the United States drought monitor. *Frontiers in Big Data*, 4, 773478. https://doi.org/10.3389/fdata.2021.773478
- Buban, M. S., Lee, T. R., & Baker, C. B. (2020). A comparison of the U.S. climate reference network precipitation data to the parameter-elevation regressions on independent slopes model (PRISM). *Journal of Hydrometeorology*, 21(10), 2391–2400. https://doi.org/10.1175/ JHM-D-19-0232.1
- Burek, P., Satoh, Y., Kahil, T., Tang, T., Greve, P., Smilovic, M., et al. (2020). Development of the community water model (CWatM v1.04)—A high-resolution hydrological model for global and regional assessment of integrated water resources management. *Geoscientific Model Development*, 13(7), 3267–3298. https://doi.org/10.5194/gmd-13-3267-2020
- Chen, H., Wang, S., Wang, Y., & Zhu, J. (2020). Probabilistic projections of hydrological droughts through convection-permitting climate simulations and multimodel hydrological predictions. *Journal of Geophysical Research: Atmospheres*, 125(15), e2020JD032914. https://doi. org/10.1029/2020JD032914

## Acknowledgments

This study was supported by NOAA's Climate Program Office's Modeling, Analysis, Predictions, and Projections Program (MAPP), Grants NA20OAR4310421 and NA17OAR4310134. This research was partially supported by the U.S. Geological Survery (USGS) Water Mission Area's Integrated Water Prediction Program (Grant number 140G0121F0357). National Center for Atmospheric Research (NCAR) is a major facility sponsored by the National Science Foundation (NSF) under Cooperative Agreement 1852977. We also acknowledge the support from the NCAR Water System Program. Any opinions, findings, conclusions, or recommendations expressed in this publication are those of the authors and do not necessarily reflect the views of the National Science Foundation. We thank Dr. Trenton W. Ford for sharing soil moisture observational data.

- Currier, W. R., Thorson, T., & Lundquist, J. D. (2017). Independent evaluation of frozen precipitation from WRF and PRISM in the Olympic mountains. *Journal of Hydrometeorology*, 18(10), 2681–2703. https://doi.org/10.1175/JHM-D-17-0026.1
- Dai, A. (2011). Characteristics and trends in various forms of the palmer drought severity index during 1900–2008. Journal of Geophysical Research, 116(D12), D12115. https://doi.org/10.1029/2010JD015541
- Daly, C., Halbleib, M., Smith, J. I., Gibson, W. P., Doggett, M. K., Taylor, G. H., et al. (2008). Physiographically sensitive mapping of climatological temperature and precipitation across the conterminous United States. *International Journal of Climatology*, 28(15), 2031–2064. https:// doi.org/10.1002/joc.1688
- Daly, C., Smith, J. I., & Olson, K. V. (2015). Mapping atmospheric moisture climatologies across the conterminous United States. *PLoS One*, *10*, e0141140. https://doi.org/10.1371/journal.pone.0141140
- Davenport, F., Husak, G., & Jayanthi, H. (2015). Simulating regional grain yield distributions to support agricultural drought risk assessment. *Applied Geography*, 63, 136–145. https://doi.org/10.1016/j.apgeog.2015.06.010
- Dierauer, J. R., Allen, D. M., & Whitfield, P. H. (2019). Snow drought risk and susceptibility in the western United States and southwestern Canada. Water Resources Research, 55(4), 3076–3091. https://doi.org/10.1029/2018WR023229
- Dormann, C. F., Elith, J., Bacher, S., Buchmann, C., Carl, G., Carré, G., et al. (2013). Collinearity: A review of methods to deal with it and a simulation study evaluating their performance. *Ecography*, 36(1), 27–46. https://doi.org/10.1111/j.1600-0587.2012.07348.x
- Enfield, D. B., Mestas-Nunez, A. M., & Trimble, P. J. (2001). The Atlantic Multidecadal Oscillation and its relationship to rainfall and river flows in the continental U.S. *Geophysical Research Letters*, 28(10), 2077–2080. https://doi.org/10.1029/2000gl012745
- Fankhauser, K., Macharia, D., Coyle, J., Kathuni, S., McNally, A., Slinski, K., & Thomas, E. (2022). Estimating groundwater use and demand in arid Kenya through assimilation of satellite data and in situ sensors with machine learning toward drought early action. *Science of the Total Environment*, 831, 154453. https://doi.org/10.1016/j.scitoteny.2022.154453
- Fontaine, M. M., Steinemann, A. C., & Hayes, M. J. (2014). State drought programs and plans: Survey of the western United States. Natural Hazards Review, 15(1), 95–99. https://doi.org/10.1061/(ASCE)NH.1527-6996.0000094
- Ford, T., & Labosier, C. F. (2014). Spatial patterns of drought persistence in the southeastern United States: Drought persistence in the southeast. International Journal of Climatology, 34(7), 2229–2240. https://doi.org/10.1002/joc.3833
- Ford, T. W., & Quiring, S. M. (2019). Comparison of contemporary in situ, model, and satellite remote sensing soil moisture with a focus on drought monitoring. *Water Resources Research*, 55(2), 1565–1582. https://doi.org/10.1029/2018WR024039
- Fung, K. F., Huang, Y. F., Koo, C. H., & Soh, Y. W. (2020). Drought forecasting: A review of modelling approaches 2007–2017. Journal of Water and Climate Change, 11(3), 771–799. https://doi.org/10.2166/wcc.2019.236
- Hao, Z., & AghaKouchak, A. (2014). A nonparametric multivariate multi-index drought monitoring framework. Journal of Hydrometeorology, 15(1), 89–101. https://doi.org/10.1175/JHM-D-12-0160.1

Hao, Z., AghaKouchak, A., Nakhjiri, N., & Farahmand, A. (2014). Global integrated drought monitoring and prediction system. Scientific Data, 1, 140001. https://doi.org/10.1038/sdata.2014.1

- Hao, Z., Hao, F., Xia, Y., Singh, V. P., Hong, Y., Shen, X., & Ouyang, W. (2016). A statistical method for categorical drought prediction based on NLDAS-2. Journal of Applied Meteorology and Climatology, 55(4), 1049–1061. https://doi.org/10.1175/JAMC-D-15-0200.1
- Hao, Z., Singh, V. P., & Xia, Y. (2018). Seasonal drought prediction: Advances, challenges, and future prospects. *Reviews of Geophysics*, 56(1), 108–141. https://doi.org/10.1002/2016RG000549
- Hao, Z., Xia, Y., Luo, L., Singh, V. P., Ouyang, W., & Hao, F. (2017). Toward a categorical drought prediction system based on U.S. drought monitor (USDM) and climate forecast. *Journal of Hydrology*, 551, 300–305. https://doi.org/10.1016/j.jhydrol.2017.06.005
- Hao, Z., Yuan, X., Xia, Y., Hao, F., & Singh, V. P. (2017). An overview of drought monitoring and prediction systems at regional and global scales. Bulletin of the American Meteorological Society, 98(9), 1879–1896. https://doi.org/10.1175/BAMS-D-15-00149.1
- Hastie, T., & Tibshirani, R. (1986). Generalized additive models. *Statistical Science*, 1(3), 297–318. https://doi.org/10.1214/ss/1177013604
- He, C., Chen, F., Abolafia-Rosenzweig, R., Ikeda, K., Liu, C., & Rasmussen, R. (2021). What causes the unobserved early-spring snowpack ablation in convection-permitting WRF modeling over Utah mountains? *Journal of Geophysical Research: Atmospheres*, 126(22), e2021JD035284. https://doi.org/10.1029/2021JD035284
- He, C., Chen, F., Barlage, M., Liu, C., Newman, A., Tang, W., et al. (2019). Can convection-permitting modeling provide decent precipitation for offline high-resolution snowpack simulations over mountains? *Journal of Geophysical Research: Atmospheres*, 124(23), 12631–12654. https://doi.org/10.1029/2019JD030823
- He, M., Russo, M., & Anderson, M. (2016). Predictability of seasonal streamflow in a changing climate in the Sierra Nevada. Climate, 4, 57. https://doi.org/10.3390/cli4040057
- Huning, L. S., & AghaKouchak, A. (2020). Global snow drought hot spots and characteristics. Proceedings of the National Academy of Sciences of the United States of America, 117(33), 19753–19759. https://doi.org/10.1073/pnas.1915921117
- Ikeda, K., Rasmussen, R., Liu, C., Newman, A., Chen, F., Barlage, M., et al. (2021). Snowfall and snowpack in the western U.S. As captured by convection permitting climate simulations: Current climate and pseudo global warming future climate. *Climate Dynamics*, 57(7–8), 2191– 2215. https://doi.org/10.1007/s00382-021-05805-w
- King, J. (2022). pdsi. GitHub. Retrieved from https://github.com/JonKing93/pdsi/releases/tag/v1.0.0
- Kirtman, B. P., Min, D., Infanti, J. M., Kinter, J. L., Paolino, D. A., Zhang, Q., et al. (2014). The North American multimodel ensemble: Phase-1 seasonal-to-interannual prediction; phase-2 toward developing intraseasonal prediction. *Bulletin of the American Meteorological Society*, 95(4), 585–601. https://doi.org/10.1175/BAMS-D-12-00050.1
- Kogan, F. N. (1997). Global drought watch from space. Bulletin of the American Meteorological Society, 78(4), 621–636. https://doi. org/10.1175/1520-0477(1997)078<0621:GDWFS>2.0.CO;2
- Koster, R. D., & Suarez, M. J. (2001). Soil moisture memory in climate models. *Journal of Hydrometeorology*, 2(6), 558–570. https://doi.org/10 .1175/1525-7541(2001)002<0558:SMMICM>2.0.CO;2
- Kuhn-Régnier, A., Voulgarakis, A., Nowack, P., Forkel, M., Prentice, I. C., & Harrison, S. P. (2021). The importance of antecedent vegetation and drought conditions as global drivers of burnt area. *Biogeosciences*, 18(12), 3861–3879. https://doi.org/10.5194/bg-18-3861-2021
- Kumar, S. V., Peters-Lidard, C. D., Mocko, D., Reichle, R., Liu, Y., Arsenault, K. R., et al. (2014). Assimilation of remotely sensed soil moisture and snow depth retrievals for drought estimation. *Journal of Hydrometeorology*, 15(6), 2446–2469. https://doi.org/10.1175/JHM-D-13-0132.1 Lakshmi, V., Piechota, T., Naravan, U., & Tang, C. (2004). Soil moisture as an indicator of weather extremes: Soil moisture climate extremes.
- .aksmit, V., Piechola, I., Narayan, U., & Tang, C. (2004). Soli moisture as an indicator of weather extremes. Soli moisture climate extremes. Geophysical Research Letters, 31(11), L11401. https://doi.org/10.1029/2004GL019930
- Littell, J. S., McKenzie, D., Peterson, D. L., & Westerling, A. L. (2009). Climate and wildfire area burned in western U.S. ecoprovinces, 1916– 2003. Ecological Applications, 19(4), 1003–1021. https://doi.org/10.1890/07-1183.1
- Littell, J. S., Peterson, D. L., Riley, K. L., Liu, Y., & Luce, C. H. (2016). A review of the relationships between drought and forest fire in the United States. *Global Change Biology*, 22(7), 2353–2369. https://doi.org/10.1111/gcb.13275

- Liu, C., Ikeda, K., Rasmussen, R., Barlage, M., Newman, A. J., Prein, A. F., et al. (2017). Continental-scale convection-permitting modeling of the current and future climate of North America. *Climate Dynamics*, 49(1–2), 71–95. https://doi.org/10.1007/s00382-016-3327-9
- Livneh, B., & Hoerling, M. P. (2016). The physics of drought in the U.S. central great plains. Journal of Climate, 29(18), 6783–6804. https://doi.org/10.1175/JCLI-D-15-0697.1
- Lobell, D. B., Hammer, G. L., McLean, G., Messina, C., Roberts, M. J., & Schlenker, W. (2013). The critical role of extreme heat for maize production in the United States. *Nature Climate Change*, 3(5), 497–501. https://doi.org/10.1038/nclimate1832
- Lobell, D. B., Roberts, M. J., Schlenker, W., Braun, N., Little, B. B., Rejesus, R. M., & Hammer, G. L. (2014). Greater sensitivity to drought accompanies maize yield increase in the U.S. midwest. *Science*, 344(6183), 516–519. https://doi.org/10.1126/science.1251423
- Lorenz, D. J., Otkin, J. A., Svoboda, M., Hain, C. R., Anderson, M. C., & Zhong, Y. (2017). Predicting the U.S. drought monitor using precipitation, soil moisture, and evapotranspiration anomalies. Part II: Intraseasonal drought intensification forecasts. *Journal of Hydrometeorology*, 18(7), 1963–1982. https://doi.org/10.1175/JHM-D-16-0067.1
- Luo, L., & Wood, E. F. (2007). Monitoring and predicting the 2007 U.S. drought. Geophysical Research Letters, 34(22), L22702. https://doi. org/10.1029/2007GL031673
- Madadgar, S., & Moradkhani, H. (2013). A Bayesian framework for probabilistic seasonal drought forecasting. Journal of Hydrometeorology, 14(6), 1685–1705. https://doi.org/10.1175/JHM-D-13-010.1
- Madadgar, S., & Moradkhani, H. (2014). Spatio-temporal drought forecasting within Bayesian networks. Journal of Hydrology, 512, 134–146. https://doi.org/10.1016/j.jhydrol.2014.02.039
- Mahanama, S. P. P., & Koster, R. D. (2003). Intercomparison of soil moisture memory in two land surface models. *Journal of Hydrometeorology*, 4(6), 1134–1146. https://doi.org/10.1175/1525-7541(2003)004<1134:IOSMMI>2.0.CO;2
- Mantua, N. J. (1999). The Pacific Decadal Oscillation. A brief overview for non-specialists. Encyclopedia of Environmental Change.
- Mathivha, F., Sigauke, C., Chikoore, H., & Odiyo, J. (2020). Short-term and medium-term drought forecasting using generalized additive models. Sustainability, 12(10), 4006. https://doi.org/10.3390/su12104006
- McCabe, G. J., Palecki, M. A., & Betancourt, J. L. (2004). Pacific and Atlantic Ocean influences on multidecadal drought frequency in the United States. Proceedings of the National Academy of Sciences of the United States of America, 101(12), 4136–4141. https://doi.org/10.1073/ pnas.0306738101
- Mo, K. C., & Schemm, J. E. (2008). Relationships between ENSO and drought over the southeastern United States. Geophysical Research Letters, 35(15), L15701. https://doi.org/10.1029/2008GL034656
- Mu, Q., Zhao, M., & Running, S. W. (2013). MODIS global terrestrial evapotranspiration (ET) product (NASA MOD16A2/A3). Algorithm Theoretical Basis Document, Collection, 5, 600.
- NOAA National Centers for Environmental Information (NCEI) U.S. (2022). Billion-dollar weather and climate disasters. Retrieved from https:// www.ncei.noaa.gov/access/billions/
- Otkin, J. A., Anderson, M. C., Hain, C., Mladenova, I. E., Basara, J. B., & Svoboda, M. (2013). Examining rapid onset drought development using the thermal infrared–based evaporative stress index. *Journal of Hydrometeorology*, 14(4), 1057–1074. https://doi.org/10.1175/ JHM-D-12-0144.1
- Otkin, J. A., Shafer, M., Svoboda, M., Wardlow, B., Anderson, M. C., Hain, C., & Basara, J. (2015). Facilitating the use of drought early warning information through interactions with agricultural stakeholders. *Bulletin of the American Meteorological Society*, 96(7), 1073–1078. https:// doi.org/10.1175/BAMS-D-14-00219.1
- Otkin, J. A., Svoboda, M., Hunt, E. D., Ford, T. W., Anderson, M. C., Hain, C., & Basara, J. B. (2018). Flash droughts: A review and assessment of the challenges imposed by rapid-onset droughts in the United States. *Bulletin of the American Meteorological Society*, 99(5), 911–919. https://doi.org/10.1175/BAMS-D-17-0149.1
- Palmer, W. C. (1965). Meteorological drought. Weather Bureau, Research Paper No. 45 (p. 58). U. S. Department of Commerce.
- Park, S., Im, J., Jang, E., & Rhee, J. (2016). Drought assessment and monitoring through blending of multi-sensor indices using machine learning approaches for different climate regions. Agricultural and Forest Meteorology, 216, 157–169. https://doi.org/10.1016/j.agrformet.2015.10.011
- Pearson, K., & Henrici, O. M. F. E. (1896). VII. Mathematical contributions to the theory of evolution—III. Regression, heredity, and panmixia. Philosophical Transactions of the Royal Society of London - Series A: Containing Papers of a Mathematical or Physical Character, 187, 253–318. https://doi.org/10.1098/rsta.1896.0007
- Prein, A. F., Langhans, W., Fosser, G., Ferrone, A., Ban, N., Goergen, K., et al. (2015). A review on regional convection-permitting climate modeling: Demonstrations, prospects, and challenges. *Reviews of Geophysics*, 53(2), 323–361. https://doi.org/10.1002/2014RG000475
- Prein, A. F., Towler, E., Ge, M., Llewellyn, D., Baker, S., Tighi, S., & Barrett, L. (2022). Sub-seasonal predictability of North American monsoon precipitation. *Geophysical Research Letters*, 49(9), e2021GL095602. https://doi.org/10.1029/2021GL095602
- Prodhan, F. A., Zhang, J., Hasan, S. S., Pangali Sharma, T. P., & Mohana, H. P. (2022). A review of machine learning methods for drought hazard monitoring and forecasting: Current research trends, challenges, and future research directions. *Environmental Modelling & Software*, 149, 105327. https://doi.org/10.1016/j.envsoft.2022.105327
- Quiring, S. M., & Kluver, D. B. (2009). Relationship between winter/spring snowfall and summer precipitation in the northern great plains of North America. Journal of Hydrometeorology, 10(5), 1203–1217. https://doi.org/10.1175/2009JHM1089.1
- R Core Team. R. (2020). A language and environment for statistical computing. R Foundation for Statistical Computing. Retrieved from https://www.R-project.org/
- Rahman, M. M., Lu, M., & Kyi, K. H. (2015). Variability of soil moisture memory for wet and dry basins. Journal of Hydrology, 523, 107–118. https://doi.org/10.1016/j.jhydrol.2015.01.033
- Rajagopalan, B., Cook, E., Lall, U., & Ray, B. K. (2000). Spatiotemporal variability of ENSO and SST teleconnections to summer drought over the United States during the twentieth century. *Journal of Climate*, 13(24), 4244–4255. https://doi.org/10.1175/1520-0442(2000)013<4244:SVOE AS>2.0.CO;2
- Rasmussen, R. M., Liu, C., Ikeda, K., Chen, F., Kim, J., Schneider, T., et al. (2023). Four-kilometer long-term regional hydroclimate reanalysis over the conterminous United States (CONUS), 1979-2020. U.S. Geological Survey Data Release. https://doi.org/10.5066/P9PHPK4F
- Rhee, J., & Im, J. (2017). Meteorological drought forecasting for ungauged areas based on machine learning: Using long-range climate forecast and remote sensing data. Agricultural and Forest Meteorology, 237–238, 105–122. https://doi.org/10.1016/j.agrformet.2017.02.011
- Ryu, J. H., Svoboda, M. D., Lenters, J. D., Tadesse, T., & Knutson, C. L. (2010). Potential extents for ENSO-driven hydrologic drought forecasts in the United States. *Climate Change*, 101(3–4), 575–597. https://doi.org/10.1007/s10584-009-9705-0
- Samaniego, L., Thober, S., Wanders, N., Pan, M., Rakovec, O., Sheffield, J., et al. (2019). Hydrological forecasts and projections for improved decision-making in the water sector in Europe. Bulletin of the American Meteorological Society, 100(12), 2451–2472. https://doi.org/10.1175/ BAMS-D-17-0274.1

- Scaff, L., Ikeda, K., Rasmussen, R., Matte, D., Prein, A. F., Liu, C., & Li, L. (2020). CONUS II, a new 20-year convection-permitting simulation over North America: How well does it performs on snowpack? In AGU fall meeting abstracts (Vol. 2020, pp. A096–A0003).
- Scanlon, B. R., Faunt, C. C., Longuevergne, L., Reedy, R. C., Alley, W. M., McGuire, V. L., & McMahon, P. B. (2012). Groundwater depletion and sustainability of irrigation in the US high plains and Central Valley. *Proceedings of the National Academy of Sciences of the United States* of America, 109(24), 9320–9325. https://doi.org/10.1073/pnas.1200311109
- Shin, C.-S., Huang, B., Dirmeyer, P. A., Halder, S., & Kumar, A. (2020). Sensitivity of U.S. drought prediction skill to land initial states. Journal of Hydrometeorology, 21(12), 2793–2811. https://doi.org/10.1175/JHM-D-20-0025.1
- Shukla, S., McNally, A., Husak, G., & Funk, C. (2014). A seasonal agricultural drought forecast system for food-insecure regions of East Africa. *Hydrology and Earth System Sciences*, 18(10), 3907–3921. https://doi.org/10.5194/hess-18-3907-2014
- Shukla, S., Steinemann, A. C., & Lettenmaier, D. P. (2011). Drought monitoring for Washington state: Indicators and applications. Journal of Hydrometeorology, 12(1), 66–83. https://doi.org/10.1175/2010JHM1307.1
- Strzepek, K., Yohe, G., Neumann, J., & Boehlert, B. (2010). Characterizing changes in drought risk for the United States from climate change. *Environmental Research Letters*, 5(4), 044012. https://doi.org/10.1088/1748-9326/5/4/044012
- Svoboda, M., LeComte, D., Hayes, M., Heim, R., Gleason, K., Angel, J., et al. (2002). The drought monitor. Bulletin of the American Meteorological Society, 83(8), 1181–1190. https://doi.org/10.1175/1520-0477-83.8.1181

Szép, I. J., Mika, J., & Dunkel, Z. (2005). Palmer drought severity index as soil moisture indicator: Physical interpretation, statistical behavior and relation to global climate. *Physics and Chemistry of the Earth, Parts A/B/C*, 30(1–3), 231–243. https://doi.org/10.1016/j.pce.2004.08.039

- Taylor, K. E. (2001). Summarizing multiple aspects of model performance in a single diagram. *Journal of Geophysical Research*, 106(D7), 7183–7192. https://doi.org/10.1029/2000JD900719
- Technical Committee on Standardization of Reference Evapotranspiration. (2005). The ASCE standardized reference evapotranspiration equation. American Society of Civil Engineers. https://doi.org/10.1061/9780784408056
- Van Loon, A. F. (2015). Hydrological drought explained. WIREs Water, 2(4), 359-392. https://doi.org/10.1002/wat2.1085
- Vergopolan, N., Chaney, N. W., Pan, M., Sheffield, J., Beck, H. E., Ferguson, C. R., et al. (2021). SMAP-HydroBlocks, a 30-m satellite-based soil moisture dataset for the conterminous US. *Scientific Data*, 8(1), 264. https://doi.org/10.1038/s41597-021-01050-2
- Vergopolan, N., Sheffield, J., Chaney, N. W., Pan, M., Beck, H. E., Ferguson, C. R., et al. (2022). High-resolution soil moisture data reveal complex multi-scale spatial variability across the United States. *Geophysical Research Letters*, 49(15), e2022GL098586. https://doi. org/10.1029/2022GL098586
- Wallander, R., Hrozencik, A., & Aillery, M. (2022). Irrigation organizations: Drought planning and response. USDA Economic Research Service. Economic Brief Number 33 Retrieved from https://www.ers.usda.gov/publications/pub-details/?pubid=102949
- Westerling, A. L., Hidalgo, H. G., Cayan, D. R., & Swetnam, T. W. (2006). Warming and earlier spring increase western US forest wildfire activity. Science, 313(5789), 940–943. https://doi.org/10.1126/science.1128834
- Wheeler, K. G., Udall, B., Wang, J., Kuhn, E., Salehabadi, H., & Schmidt, J. C. (2022). What will it take to stabilize the Colorado River? Science, 377(6604), 373–375. https://doi.org/10.1126/science.abo4452
- Williams, A. P., Cook, B. I., & Smerdon, J. E. (2022). Rapid intensification of the emerging southwestern North American megadrought in 2020–2021. Nature Climate Change, 12(3), 232–234. https://doi.org/10.1038/s41558-022-01290-z
- Williams, A. P., Cook, E. R., Smerdon, J. E., Cook, B. I., Abatzoglou, J. T., Bolles, K., et al. (2020). Large contribution from anthropogenic warming to an emerging North American megadrought. *Science*, 368(6488), 314–318. https://doi.org/10.1126/science.aaz9600
- Wlostowski, A. N., Jennings, K. S., Bash, R. E., Burkhardt, J., Wobus, C. W., & Aggett, G. (2022). Dry landscapes and parched economies: A review of how drought impacts nonagricultural socioeconomic sectors in the US Intermountain West. WIREs Water, 9(1). https://doi. org/10.1002/wat2.1571
- Wood, S. N. (2011). Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models. *Journal of the Royal Statistical Society*, 73(1), 3–36. https://doi.org/10.1111/j.1467-9868.2010.00749.x
- Xia, Y., Ek, M., Wei, H., & Meng, J. (2012). Comparative analysis of relationships between NLDAS-2 forcings and model outputs. *Hydrological Processes*, 26(3), 467–474. https://doi.org/10.1002/hyp.8240
- Xia, Y., Ek, M. B., Peters-Lidard, C. D., Mocko, D., Svoboda, M., Sheffield, J., & Wood, E. F. (2014). Application of USDM statistics in NLDAS-2: Optimal blended NLDAS drought index over the continental United States. *Journal of Geophysical Research: Atmospheres*, 119(6), 2947– 2965. https://doi.org/10.1002/2013JD020994
- Xia, Y., Mitchell, K., Ek, M., Cosgrove, B., Sheffield, J., Luo, L., et al. (2012). Continental-scale water and energy flux analysis and validation for North American land data assimilation system project phase 2 (NLDAS-2): 2. Validation of model-simulated streamflow. *Journal of Geophysical Research*, 117(D3), D03110. https://doi.org/10.1029/2011JD016051
- Yan, H., Moradkhani, H., & Zarekarizi, M. (2017). A probabilistic drought forecasting framework: A combined dynamical and statistical approach. Journal of Hydrology, 548, 291–304. https://doi.org/10.1016/j.jhydrol.2017.03.004
- Yoder, R. E., Odhiambo, L. O., & Wright, W. C. (2005). Effects of vapor-pressure deficit and net-irradiance calculation methods on accuracy of standardized Penman-Monteith equation in a humid climate. *Journal of Irrigation and Drainage Engineering*, 131(3), 228–237. https://doi. org/10.1061/(ASCE)0733-9437(2005)131:3(228)
- Yoon, J.-H., Mo, K., & Wood, E. F. (2012). Dynamic-model-based seasonal prediction of meteorological drought over the contiguous United States. *Journal of Hydrometeorology*, 13(2), 463–482. https://doi.org/10.1175/JHM-D-11-038.1
- Zeng, X., Broxton, P., & Dawson, N. (2018). Snowpack change from 1982 to 2016 over conterminous United States. Geophysical Research Letters, 45(23), 12940–12947. https://doi.org/10.1029/2018gl079621
- Zhu, Y., Wang, W., Singh, V. P., & Liu, Y. (2016). Combined use of meteorological drought indices at multi-time scales for improving hydrological drought detection. *Science of the Total Environment*, 571, 1058–1068. https://doi.org/10.1016/j.scitotenv.2016.07.096
- Zipper, S. C., Qiu, J., & Kucharik, C. J. (2016). Drought effects on US maize and soybean production: Spatiotemporal patterns and historical changes. Environmental Research Letters, 11(9), 094021. https://doi.org/10.1088/1748-9326/11/9/094021